Author's response to reviews

Title: Measurement properties of the Minimal Insomnia Symptom Scale (MISS) in an elderly population in Sweden.

Authors:

Amanda Hellström (amanda.hellstrom@bth.se)
Peter Hagell (peter.hagell@med.lu.se)
Cecilia Fagerström (cecilia.fagerstrom@bth.se)
Ania Willman (ania.willman@bth.se)

Version: 5 Date: 4 November 2010

Author's response to reviews: see over
Cover letter

This is a point by point description of the revisions made in manuscript, tables and figures, as requested by the editorial production team.

1. In the “Authors’ contributions”-passage a sentence has been added, declaring that all of the authors have read and approved of the manuscript.

2. Track changes have been removed from the text.

3. Visible, vertical lines in the tables have been removed.

4. Figure 1 and Figure 2 have been cropped once more, following the instructions and recommendations of the BMC Geriatrics.

5. The text has been read in its entirety and checked for errors.